Author's response to reviews

Title: A modified echocardiographic protocol with intrinsic plausibility control to determine intraventricular asynchrony based on TDI and TSI

Authors:

Henryk Dreger (henryk.dreger@charite.de)
Adrian C. Borges (adrian.borges@charite.de)
Bruno Ismer (bruno.ismer@med.uni-rostock.de)
Sebastian Schattke (sebastian.schattke@charite.de)
Berthold Stegemann (berthold.stegemann@tele2.de)
Gert Baumann (gert.baumann@charite.de)
Christoph Melzer (christoph.melzer@charite.de)

Version: 4 Date: 16 September 2009

Author's response to reviews:

We have made all requested formatting changes and hope that our manuscript now meets the requirements for publication in Cardiovascular Ultrasound.